Author’s response to reviews

Title: Factors associated with first-line antiretroviral treatment failure in adult HIV-positive patients: a case-control study from Ethiopia

Authors:
Yihienew Mequanint Bezabih (myihienew@gmail.com;yihienew.bezabih@oniris-nantes.fr)
Fekadu Beyene (bsfekadu@yahoo.com)
Woldesellassie Bezabhe (Woldesellassie.Bezabhe@utas.edu.au)

Version: 2 Date: 30 May 2019

Author’s response to reviews:

I would like to extend my deepest gratitude for your reviews. We revised the manuscript based on your comment. We highlighted our response in the manuscript and replied to the editor comments below.

Editor Comments:

1) We assessed the modifications made to your manuscript after the contributions of the reviewers, and we believe that the quality of the paper has improved. We appreciate the correct response to each of the suggested points and the modifications made to your manuscript. I want to point out a small change that should be made, and it is the replacement of the phrase "cluster of differentiation 4" by merely CD4 lymphocyte. The experience of using this abbreviation makes it unnecessary to detail what it means.

We replaced the phrase "cluster of differentiation 4" by CD4 lymphocyte.